Author's response to reviews

Title: Two novel missense substitutions in the VSX1 gene: Clinical and Genetic analysis of families with Keratoconus from India

Authors:

Rohit Shetty Dr (drrohitshetty@yahoo.com)
Rudy Nuijts Dr (rudy.nuijts@mumc.nl)
Soumya Ganesh Nanaiah Dr (soumya.kaveri@yahoo.com)
Venkata Ramana Anandula Dr (ramanagene@gmail.com)
Arkasubhra Ghosh Dr (arkasubhra@narayananethralaya.com)
Chaitra Jayadev Dr (drchaitra@hotmail.com)
Natasha Pahuja Dr (dr.natashapahuja@gmail.com)
Kumaramanic Govindasamy Dr (gkumarmvel@gmail.com)
Nallathambi Jeyabalan Dr (drnallathambi@narayananethralaya.com)

Version: 3 Date: 5 December 2014

Author's response to reviews: see over
Dear Editor,

We submit the revised version of the manuscript entitled “Two novel missense substitutions in the VSX1 gene: Clinical and Genetic analysis of families with Keratoconus from India”. We sincerely appreciate the review provided by the Journal and the invaluable comments from the reviewers. This has immensely improved the quality of our manuscript. In the revised manuscript, we have made the necessary revisions as per the comments raised. Below are our replies to the reviewer’s comments. Our revisions are in red. All the modifications and corrections in the manuscript text have been highlighted in yellow. As suggested by the editor, a sentence about the Ethics Committee approval has been added with the Approval Number.

We hope our revised version meets the high standards of the journal. Thank you for your kind consideration.

Best regards,
Sincerely

Dr Nallathambi
GROW Research Laboratory,
Narayana Nethralaya Foundation,
Bangalore, India